Author's response to reviews

Title: The validity of the non-exercise activity thermogenesis questionnaire evaluated by objectively measured daily physical activity by the triaxial accelerometer

Authors:

Hidetaka Hamasaki (hhamasaki78@gmail.com)
Hidekatsu Yanai (dyanai@hospk.ncgm.go.jp)
Masafumi Kakei (mkakei@iichi.ac.jp)
Mitsuhiko Noda (mnoda@hosp.ncgm.go.jp)
Osamu Ezaki (ezaki@swu.ac.jp)

Version: 2
Date: 11 March 2014

Author's response to reviews:

The Editor
BMC Complementary & Alternative Medicine

Dear Sirs,

We will submit our MS entitled “The validity of the non-exercise activity thermogenesis questionnaire evaluated by objectively measured daily physical activity by the triaxial accelerometer (abstract 194 words, text 1,504 words, 22 references, one table and 3 figures) to BMC Complementary & Alternative Medicine.

Hamasaki H (the first author), Yanai H (corresponding author) and Ezaki O designed the study. Hamasaki H, Yanai H and Noda M performed the study. Hamasaki H, Yanai H and Kakei M analyzed the data. Hamasaki H and Yanai wrote the paper. All authors have read, revised and approved the final version of our MS.

Our MS is not under consideration elsewhere, and none of our MS’s contents have been previously published. All authors have no conflict of interests about our MS. We will appreciate your consideration of our MS for publication in the section of “Research Article” in your journal.

Sincerely yours,

Hidekatsu Yanai, MD, PhD, FACP
Director
Department of Internal Medicine,
National Center for Global Health and Medicine Kohnodai Hospital